Author's response to reviews

Title: Instrument development, data collection and characteristics of practices, staff and measures in the Improving Quality of Care in Diabetes (iQuaD) Study

Authors:

Martin P Eccles (martin.eccles@ncl.ac.uk)
Susan Hrisos (susan.hrisos@ncl.ac.uk)
Jill J Francis (j.francis@abdn.ac.uk)
Elaine Stamp (elaine.stamp@ncl.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)
Gillian Hawthorne (gillian.hawthorne@newcastle-pct.nhs.uk)
Nick Steen (nick.steen@ncl.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Marko Elovainio (marko.elovainio@thl.fi)
Justin Presseau (justin.presseau@ncl.ac.uk)
Margaret Hunter (gwhunter@btinternet.com)

Version: 4 Date: 7 June 2011

Author's response to reviews:

Dear Production Team,

In response to your email required changes I have addressed the following:

Major revisions (we require the author to make these changes)

Consent: if applicable, please include a consent statement including the following information. Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.

NOT APPLICABLE

Tables: tables should be legible on a standard page. If you cannot re-orientate the table to fit within a page, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

LANDSCAPE TABLE NOW PORTRAIT

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
I COULD NOT FIND ANY INSTANCES OF THIS.

Tables- please remove the visible vertical lines from your tables.

DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE